SPIRE-RAL-MOM-001157

Notes on project managers teleconference 28/01/02

Chair: Eric Sawyer

Present:

Lionel Duband	Х
Ben Parvin	
Berend Winter	
Chris Brockley-Blatt	X
Matt Griffin	
Peter Hargrave	Х
Jerry Lilienthal	Х
Joe Taylor	Х
Ian Pain	
Don Peterson	Х
Bruce Swinyard	
Ken King	
Dominique Pouliquen	
Gary Parks	
Jean-Louis Auguerres	
Doug Griffin	
Dave Smith	
Eric Clark	
John Delderfield	
Didier Ferrrand	
Riccardo Cerulli-Irelli	
Renato Orfei	
Brian Stobie	
Fredericke Pinsard	

General

Please inform the project the date of your latest schedule. DDR documents on Livelink, subsystem managers to check and update if necessary. IBDR documents required very soon.

Subsystem reports

IFSI (Riccardo) – DPU Not present

CEA Grenoble (Lionel) – Cooler Cooler assembly continues.

Final soldering to be done. Procedure for correcting distortion identified. FM design will be modified in a similar manner.

USK. (Don) - Shutter Funding still in flux, but not stopping work yet.

Cardiff. (Peter) Filters, calibrators PCAL, final drawings nearly complete. SCAL, STM and CQM components in manufacture. Filters, BDA support rings complete, testing to start soon. Filters in manufacture. 300mK supports being assembled (Mk 1) 300mk supports being manufactured (Mk 2)

ATC (Ian) - BSM Not present,

JPL (Jerry) - DetectorsThermal cycling dewer has arrived.BDA being assembled, available end this week.Problems with tensioning of Kevlar fibres, some new tooling required.Harness definition doc need signing to maintain JPL schedule.FMEC should be ready at the end of the month.

MSSL (Chris) – Structure Still waiting for harness details from JPL and LAM. Order for optical bench and cover is out.

LAM. (Dominique) - Mirrors, SMEC Not present

SAp (Jean-Louis) - DRCU Not present

RAL, (Dave) - AIT Not present

No.	Action	Who	Due date	Response/Comment	Status